EO-1 Weekly Status Week of April 22 – April 28, 2010
Day of Year 112 - 118
Mission Day 3449 - 3456
Earth Observing One (EO-1) General.
Scheduled 143 science Data Collection Events (DCEs) past week
Proposal sent to Headquarters for funds to support AIST SensorWeb sustaining efforts for CEOS and GEO activities. 

EO-1 imaged the Gulf of Mexico oil spill on April 25.  The ALI image is shown below with waves clearly visible that have given the oil slick a frothy appearance.
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INSTRUMENTS

All instruments operated nominally this week 
· Performed instrument calibration

· Conducted Hyperion solar calibration on April 27 at 10:42z

· Performed instrument decontamination cycles

· Conducted Hyperion deicing on April 26 from 00:25z to 15:15z

· Conducted ALI outgassing on April 26 from 00:35z to 15:25z

EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from the Solid-State-Recorders.  Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

On Thursday, April 22, there was an ESTO AIST 2008 SensorWeb 3G collaborators teleconference.  

Participants:  Dan Mandl, Stu Frye, Rob Sohlberg, Pat Cappelaere, Bogdan Oaida, Vuong Ly, and Joe Young.

Notes from this teleconference are as follows:

1. Bogdan, Oaida will present the overall HyspIRI architecture at the GSFC HyspIRI Science Symposium on May 4.

2. Dan Mandl is working on producing a new HyspIRI IPM Quad chart that will show a compelling need for the IPM’s capability to downlink processed or raw science data with low latency.

3. Dan has been in contact with a GEO-CAPE mission person, Antonio Menino, about the need and use of IPM on that mission.  
4. Pat Cappelaere and Rob Sohlberg are to meet and discuss what sciences have compelling need for use of IPM/DB, i.e., what science will be compromised or not be done if there is no IPM on Hyperion.
5. Pat has suggested use of Radio Card technology for UHF direct downlink of data but it is not a proven technology.  The advantage of Radio Card is that its use will reduce the mass and power below that for Direct Broadcast such that the HyspIRI budget allocations are met.  The disadvantage is that HyspIRI excludes use any unproven technology.
6. Dan is to ensure that JPL’s HyspIRI Science Symposium presentations, related to the IPM are consistent with ours.

On Tuesday, April 27, there was a Namibia Flood/Disease SensorWeb collaborators teleconference.  

Participants:  Dan Mandl, Stu Frye, Joerg Szarzynski, Joachim Post, Felix Kogan, Lenny Roytman, Serhiy Skakun, and Joe Young.

Notes from this teleconference are as follows: 

1. Joerg Szarzynski suggested that the proposal for implementing the full Namibia Flood/Disease SensorWeb Project should be presented at the Donor’s Conference in June but with a follow-on presentation of a proposal to perform an initial smaller portion to start the effort.  This suggestion was accepted by the team.
2. Joerg emphasized the need to finalize the proposal as soon as possible.  There is the need to discuss this issue with Guido van Langenhove to more fully understand the total process of how the proposal will be reviewed.
3. Joe Young was assigned the action to assemble all the proposal inputs into a single document and make appropriate changes to produce a first draft formal product.

4. Discussion about Guido having a worthwhile discussion with Sophia Bettencourt, from the World Bank about the SensorWeb Pilot Program and an effort to be made to have her attend the Donor’s Conference along with the World Bank designated person Jean-Christophe Carrett.

5. The team agreed that the full proposal needs an Executive Summary.
6. Stu Frye has been delivering 2-3 new EO-1 images per week of flooded regions in Namibia to pertinent hydrologists for their analysis.  One of the images has appeared in a Namibia newspaper with a NASA/EO-1 credit given. 
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Continued nominal spacecraft state of health

· No change from last weekly report.
Mission Planning

· Continuing testing of ASIST SCP to replace legacy CMS mission planning software

· Performing testing of text-based output summarizing loads in parallel with testing of hex command loads

· Creating RTS files based on spacecraft RTS dumps and CMS RTS files

· Tested SCP from lunar calibration on March 30

Flight Dynamics
· Discovered a problem in the flight dynamics that made steady increases in the range rate and RMS values starting after April 16

· Determined the cause of the these increases as the gravity volume box and surface area in STK

· Reset these values to their previous value and conducted flight dynamics with normal values in both range rate and RMS

Trending
· Troubleshooting reported data corruption on ITPS computer hard disk(s)

· Continuing testing of suspected problem program to verify it is the cause

· Researched the increasing IRU driver voltage 

· Determined the increase has been a steady increase over the life of the mission

System Administration
· ITPS trending computer continued:

· No sign of corrupted data since the Symantec weekly scan was disabled.

· Continued the upgrade process for the 4th string ASIST workstation (also used as the backup data processing workstation).

· Configured certificate based SSH between this computer and two support computers at JPL.

· Corrected the missing user shortcuts by copying over the ASIST_user_kshrc file from another computer.

· Added the window list application to the task bar.

· The computer has completed the initial FOT testing but is still waiting to be verified by the EO-1 Tech Engineer.

· Addressed the issue with the legacy mission planning and flight dynamic computers sending emails to an old computer that is no longer in use.

· Edited scripts to replace occurrences of retired email addresses with current ones.

· Edited the mail alias file with up to date addresses.

· Adjusted the account lock settings on the new data processing computers.

· Responded to IONet firewall rules that had reached their expiration date.

GROUND AND SPACE NETWORK

Station Downtimes

· Received no notification of station downtimes

Operational Discrepancies
· Observed no operational discrepancies
UPCOMING EVENTS

· Nominal Lunar calibration to occur on April 28 during the 00:23:49z umbra

· Next inclination maneuvers to be on June 15 and June 17 expected to be 200s burns (to be verified during burn planning)

Imagery Status

Scenes and Engineering Cals planned for week of April 22 – April 28, 2010      143
Total scenes and engineering calibrations planned for entire mission – approximately 53,007
Total Scenes:  ALI scenes in the Level 0 archive              48,534                (as of April 28, 2010)

                         Hyperion scenes in the Level 0 archive    48,287                
Publications and Presentations Status 

430 publications 

295 presentations

53 articles and press releases
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